Author's response to reviews

Title: Redesign of a Computerized Clinical Reminder for Colorectal Cancer Screening: A Human-Computer Interaction Evaluation

Authors:

    Jason J Saleem (Jason.Saleem@va.gov)
    David A Haggstrom (dahaggst@iupui.edu)
    Laura G Militello (LMilitello@applieddecisionscience.com)
    Mindy Flanagan (MFlanagan@iupui.edu)
    Chris L Kiess (ckiess@iupui.edu)
    Nicole Arbuckle (Nicole.Arbuckle@udri.udayton.edu)
    Bradley N Doebbeling (BDoebbel@iupui.edu)

Version: 3 Date: 25 November 2011

Author's response to reviews:

We have completed all of the major and minor formatting revisions requested below:

----------------------------------------
Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Tables: All tables must be placed in multiple columns.

Additional files: Please remove additional file 3

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the author list in the manuscript

Affiliations: Please do not write links as ranges instead list all links to the affiliations 1-4 should be written 1,2,3 and 4

Background: Please change the title "Introduction" to "Background".

Tables: please remove the visible vertical lines from your tables

Tables: please note that we are unable to correctly display merged cells where
the merged cell crosses rows: please re-layout your table without these merged elements

Additional files: Please change the title "Additional material" to "Additional files"

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

All of the above requested formatting revisions have now been completed.

Kind regards,
Jason Saleem